Author’s response to reviews

Title: End-of-life care in Germany: Study design, methods and first results of the EPACS study (Establishment of Hospice and Palliative Care Services in Germany).

Authors:

Luis Carlos Escobar Pinzón (escobar@uni-mainz.de)
Eva Münster (eva.muenster@uni-mainz.de)
Sabine Fischbeck (fischbec@uni-mainz.de)
Michael Unrath (michael.unrath@uni-vechta.de)
Matthias Claus (clausma@uni-mainz.de)
Tanja Martini (tanja.martini@uni-mainz.de)
Martin Weber (martin.weber@ukmainz.de)

Version: 4 Date: 20 July 2010

Author’s response to reviews: see over
Mainz, 20 July 2010

Dear Editorial Production Team,

Thank you very much for accepting our paper for publication.

We changed the final manuscript as follows:

Affiliations:

1Medical Center of the Johannes Gutenberg-University of Mainz, Institute of Occupational, Social and Environmental Medicine, Obere Zahlbacher Straße 67, 55131 Mainz, Germany
2Medical Center of the Johannes Gutenberg-University of Mainz, Clinic of Psychosomatic Medicine and Psychotherapy, Medical Psychology and Medical Sociology, Saarstraße 21, 55099 Mainz, Germany
3Medical Center of the Johannes Gutenberg-University of Mainz, Interdisciplinary Palliative Care Unit, Langenbeckstraße 1, 55131 Mainz, Germany

Authors’ contributions:

LCEP participated in the design and coordination of the study, interpreted the data, and performed the drafting of the manuscript. EM conceived of the study, participated in its design and coordination, interpreted the data, and performed the drafting of the manuscript.
SF participated in the design and coordination of the study, developed its questionnaire, and critically revised the manuscript. MU participated in the drafting of the manuscript, performed the statistical analysis, and interpreted the data. MC and TM participated in the statistical analysis, and revised the final draft of the manuscript. MW conceived of the study, participated in its design and coordination, interpreted the data, and critically revised the manuscript. All of the six authors were equally involved in reading and approving the final manuscript.

Table 3
We found and changes minor errors in table 3.

We further removed all highlighting from the manuscript, removed the key words and changed the (few) underlined parts to bold. Regarding the image files, we removed the legends and included them into the figure list after the references. We additionally cropped the .ppt image files according to the suggestions on your web page:
http://www.biomedcentral.com/info/ifora/figures

Yours sincerely,

Dr. Luis Carlos Escobar Pinzón (for all authors)